Contents lists available at ScienceDirect



journal homepage: www.elsevier.com/locate/eist



Megaprojects as niches of sociotechnical transitions: The case of digitalization in UK construction

Dr Eleni Papadonikolaki^{a,*}, Dr Bethan Morgan^b, Dr George Papachristos^c

^a University College London, 1-19 Torrington Place, London WC1E 7HB, United Kingdom

^b Digital Outlook Partnership, London, United Kingdom

^c Eindhoven Univerity of Technology, Eindhoven, Netherlands

ARTICLE INFO

Keywords: Aggregation Sociotechnical transitions Megaprojects Infrastructure Digital

ABSTRACT

Transitions are processes of systemic change where niches peripheral to a sociotechnical regime accumulate momentum, scale up and eventually transform its core. In contrast to this dominant narrative in transitions research, infrastructure systems exhibit the reverse process as change propagates from the regime core to its periphery. We explore this under-researched process in the case of digitalization in UK construction. We analyse six UK megaprojects that span more than 30 years as a single longitudinal embedded case, and show how the adoption of digital technologies driven by regime incumbents, seeds the processes of technology adaptation, aggregation, and system transformation. The adoption of digital technologies by incumbents is necessary to cope with megaproject scale and scope. Their adaptation to technology instigates organizational level change that starts at the regime core, accumulates with each project and makes these changes ripple across the industry and transform it.

1. Introduction

Sociotechnical transitions are evolutionary processes of systemic change at the organizational field level that alternate between stability and change (Geels et al., 2017 Smith et al., 2010). Their study has developed into a substantial research program (Köhler et al., 2019), which describes a diverse set of radical innovations emerging in protected niches, then developing along different directions with some system wide impact (Geels, 2020, Geels and Raven, 2006, Geels and Schot, 2007, Kemp et al., 1998, Raven and Geels, 2010, Schot and Geels, 2008, 2007, Smith and Raven, 2012). In their development trajectory, sociotechnical niches enter progressively larger market segments and may fundamentally transform sociotechnical systems, i.e. legacy technologies, incumbent actors and the core rules of the regime that underlie system stability and reproduction (Fuenfschilling and Truffer, 2014; Geels, 2011).

DiMaggio and Powell (1983, p. 148) refer to the organizational field as a unit of analysis, "a recognized area of institutional life: key suppliers, consumers of resources and products, regulatory agencies and other organizations that produce similar products and services". System change proceeds mostly from small, peripheral niches, or other regimes, to the core of the organizational field (Fuenfschilling and Truffer, 2014). The implication is that small scale options have received far more attention than large scale technological options in transitions literature (Geels, 2020; Kanger et al., 2020; Sovacool and Geels, 2021; Turnheim and Geels, 2019). Nevertheless, infrastructure systems delivered through megaprojects can also be important for transitions (Sovacool and Geels, 2021). Megaprojects, generally cost more than US\$ 1 billion (Flyvbjerg, 2017, 2014; Flyvbjerg et al., 2003) and become increasingly relevant

* Corresponding author.

E-mail address: e.papadonikolaki@ucl.ac.uk (D.E. Papadonikolaki).

https://doi.org/10.1016/j.eist.2023.100728

Received 26 May 2022; Received in revised form 30 March 2023; Accepted 12 April 2023

Available online 17 April 2023





^{2210-4224/© 2023} The Authors. Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

as their number grows (Flyvbjerg, 2016; Flyvbjerg et al., 2003; Woetzel et al., 2016). They include offshore wind farms (Brookes et al., 2017; Diaz and Guedes Soares, 2020), hydro-power (Currie et al., 2021), airports and high-speed railways (Denicol et al., 2021; Esposito et al., 2021) and can have a big impact on society and the environment (Biesenthal et al., 2018). Megaprojects could be seen as evidence of an acceleration phase in transitions that follow initial experimentation in diverse, small-scale niches.

This paper heeds the call for transitions research to engage with megaprojects (Sovacool and Geels, 2021). It engages with project management (PM), which is an expanding subfield of management and organization studies (Hornstein, 2015; Morris, 2012; Söderlund, 2011). Three observations substantiate the direction we take. First, core pillars of transitions frameworks and research i.e. actors, technologies and institutions, multi-level theorization and structuration theory (Geels, 2010, 2004), overlap with core pillars for research on change processes in project-based industries: multi-level theorization of change (Gann and Salter, 2000; Lundin et al., 2015; Söderlund and Sydow, 2019; Sydow and Braun, 2018; Winch, 1998), and structuration theory (Bresnen et al., 2005; Floricel et al., 2014; Sydow and Windeler, 2020; Windeler and Sydow, 2001). For transitions and project management research, structuration provides the theoretical bedrock to analyse the duality of agency and structure (Farjoun, 2010). Second, PM represents an increasing share of organizational life (Davies and Brady, 2000; Hobday, 2000; Keegan and Turner, 2002; Midler, 1995; Sydow et al., 2004; Turner, 1999; Whitley, 2006) and organizational change processes (Gareis, 2010; Gareis and Huemann, 2008; Lundin et al., 2015; Martinsuo and Hoverfält, 2018; Müller et al., 2016; Vuorinen and Martinsuo, 2018). Organizations with their structures and practices navigate the institutional landscape and engage in project-based organizing and learning (Sydow and Windeler, 2020). This is important as transitions are said to take place at the organizational field level (Geels and Schot, 2007). Third, organizational stability is a precondition for temporary organizing and project-driven change (Farjoun, 2010). Therefore, project success is best analysed in the context of a series of projects that organizations participate in, that are embedded in broader institutional processes (Dille and Söderlund, 2011; Engwall, 2003; Sense and Antoni, 2003), just as it is done with niches (Geels and Raven, 2006; Schot and Geels, 2008).

On the basis of these observations, this paper combines transitions, PM and megaprojects approaches. It analyses of six megaprojects as a single longitudinal embedded case, and studies their system wide-impact on digitalization of UK construction (Davies et al., 2009). The megaproject scale raises tensions amongst actors as the risks they face are high and magnifies industry challenges of fragmentation, coordination, communication, project design and delivery, that need to be addressed (Egan, 1998; Latham, 1994). The solution to the challenges UK construction incumbents face comes from the adoption and adaptation of digital technologies from US defence and aerospace industry (Chasen, 1981; Llewelyn, 1989; Morris, 2013). Incumbents adopt and adapt technology to meet the broad scope, high-quality requirements of megaprojects, so they are lead users of digital technology (Bogers and Bastian, 2010; Von Hippel, 1986). The digitalization process is not marked by the entry of new firms or the decline of large incumbent firms but new entries of Information and Communication Technology (ICT) firms that supply design software packages. The megaproject case does not conform to a bottom-up transition pattern, but change begins with incumbents at the core of the organizational field.

Subsequent cumulative interactions between project actors and ICT suppliers change the project-related and organizational practices they help constitute (Harty, 2005; Harty and Whyte, 2010; Morgan, 2019). The reorganization of practices within and between partners and suppliers drives the coevolution of organizations, competences and technological innovation (Ciarli et al., 2021; Leonard-Barton, 1988; Orlikowski, 1992). Some incumbents participate in subsequent megaprojects, so the co-adaptation of innovations and organizations extends to other industry actors and - along with the emergence of industry standards - transforms organizational structure and actor practices towards more integrated collaboration. This paper therefore addresses the following two Research Questions (RO):

1 How have digital technology outcomes aggregated and produced a system-wide transformational impact on the UK construction?

2 How are digital technologies in megaprojects shape and shaped by sociotechnical transitions?

The remainder of this paper is organised as follows. Section 2 introduces the theoretical background to the study and traces the correspondence of niches to projects. Section 3 summarizes data and methods used in our research. Section 4 presents the data of the case study of digital innovation in UK construction and its analysis to examine the study findings. Section 5 discusses the paper's findings with implications on the cross over between projects and transitions and Section 6 concludes the study.

2. Theoretical background

2.1. Sociotechnical transitions: landscape, regime and niches

The Multiple-Level Perspective (MLP) is the predominant transitions framework where actors and institutions interact through different rules and induce transitions (Geels, 2004; Geels and Schot, 2007; Papachristos et al., 2013). Geels et al. (2016) distinguish between: (1) actors and social groups, (2) rules and institutions, and (3) technologies that interact dynamically. Sociotechnical transitions research, then, focuses on system interconnections and social group dynamics that influence periods of system stability and change. The sociotechnical regime is the central concept for the analysis of incumbent actors and activities that reproduce or change system elements. Actors form interdependent social groups, each with its own regime or set of rules. These rules align and coordinate the activities of actor groups that modulate system inertia and change (Geels et al., 2016; Geels and Schot, 2007). The sociotechnical regime, then, is the inter-regime rule set that aligns and coordinates intergroup activities that shape sociotechnical trajectories. The sociotechnical landscape is a wider canvas of macro influences that reinforce or stem niche and regime dynamics (Geels, 2011; Rip and Kemp, 1998).

Transitions are non-linear processes at the organizational field that result from the interactions of (1) niches, where innovations develop, (2) sociotechnical regimes of established practices and associated rules and (3) long term landscape trends (Geels, 2002; Rip and Kemp, 1998). Transitions have been conceptualised into four stylised pathways: 'substitution', 'transformation', 'reconfiguration', 'de-alignment and re-alignment' (Geels et al., 2016). It is possible that a transition will adhere to one, or shift from one pathway to another. Transitions are initiated when the sociotechnical regime is destabilised through niche innovations, internal regime tensions, landscape trends that put pressure on the focal regime (e.g. climate, economic, cultural, demographic and other), and external influences from other systems, regimes or niches (Geels and Schot, 2007; Papachristos et al., 2013; Raven, 2007; Rosenbloom, 2020). A transition proceeds through a process of accumulation of local, peripheral niche activities and subsequently enter increasingly larger (market) niches at the centre of the organizational field (Geels, 2002; Geels and Schot, 2007).

2.2. Niche development and empowerment

Niches can act as peripheral, 'protective spaces' where selection criteria are more favourable to new technologies than in the mainstream markets that operate under the dominant regime (Levinthal, 1998; Schot and Geels, 2007). In these shielded markets, the interaction of users with producers, gives rise to mutual learning and expectation articulation processes (Hoogma et al., 2002; Kemp et al., 1998). Niche *nurturing* involves articulation of actor *expectations* and visions, development of social *networks*, and *learning* processes (Schot and Geels, 2008). Learning processes contribute to niche development and involve first and second order processes. In the early stages of niche development, technology performance criteria are unclear, so technological competition is based on future expectations rather than current performance (Budde and Konrad, 2019; Klepper, 1997; Rosenberg, 1976).

Expectations and visions motivate, engage and align actors to commit their resources towards promising technological fields (Konrad, 2006), and guide niche learning and development (Geels and Raven, 2006). Niche *empowerment* is integral to transitions because it can influence the regime. Two empowerment processes modulate the interface between niche(s) and the regime (Raven, 2007; Smith et al., 2010; Smith and Raven, 2012): *fit and conform* and *stretch and transform*. Niches are empowered through *stretch and transformation*, by presenting a realistic alternative to problems, instabilities and tensions the regime experiences, such that tech implementation and institutionalization of niche practices becomes accepted by regime actors. In *fit and conform*, niches are empowered when they improve along established performance dimensions.

The process of technology implementation involves the mutual adaptation of technology and organizations (Leonard-Barton, 1988; Orlikowski, 1992; Tyre and Orlikowski, 1994). Market niches materialize as the product of organizational action (Astley, 1985; McKelvey, 1982). Thus, markets, user preferences and competences may need to be co-constructed with new technologies (Leonard-Barton, 1988; Oudshoorn and Pinch, 2003), to meet various attributes and selection criteria, such as price/performance, and the minimum functionality threshold for a technology in an application domain (Adner and Levinthal, 2001). Technology may have a more aggregate effect at the organizational field level, if technological functionality increases, or its cost is reduced and subsequently reaches larger, mainstream niches and becomes a general-purpose technology (Rosenberg, 1976).

2.3. Niche aggregation and structuration from local to the global level

The process of niche aggregation involves four phases (Geels and Deuten, 2006). In the first (local) phase, new technologies emerge or are introduced through technology absorption and relatively independent new entrants in local practices who create local knowledge for their own purposes. In the second (inter-local) phase, technological knowledge flows initiate in actor networks. Learning and accumulation of experiences has an inter-organizational but still local niche character through people, professional societies and industry associations that stimulate and facilitate the production and circulation of technical knowledge through interaction between ICT firms, users and regulators. When innovation experience is transferred from one project to another then general lessons are developed, and local knowledge is gradually absorbed into generic knowledge (Fleck, 1994).

In the third (trans-local) phase, knowledge production and circulation increase and acquire a global scope as groups of firms coalesce around their collective interests. In this phase a knowledge infrastructure emerges with dedicated journals and conferences for knowledge circulation, and with intermediary actors that perform dedicated knowledge activities at the global level to create, standardise, and distribute knowledge (Kivimaa et al., 2018; Schmidt and Werle, 1998). In the fourth (global/cosmopolitan) phase, dominant cognitive rules and a global stock of knowledge, technology artefacts and standards become established and shape local-level activities. Technological development thus proceeds simultaneously at local and global levels (Geels and Deuten, 2006; Geels and Raven, 2006; Schot and Geels, 2008). Therefore, the aggregation of peripheral niches produces generalizable knowledge through shared cognitive rules, structures and standards (the most tangible and codified outcomes of aggregation), and a system wide impact that catalyses a transition (Geels and Deuten, 2006; Geels and Raven, 2006; Raven and Geels, 2010; Schot and Geels, 2008).

2.4. Megaprojects as sociotechnical niches

A motivation to view megaprojects as sociotechnical niches is the argument made in PM literature that projects must be considered as sociotechnical endeavours embedded in complex institutional settings (Biesenthal et al., 2018). Modern infrastructures in transport, energy, and telecommunications are delivered through megaprojects, with investment exceeding \$1 billion. Megaprojects are capital intensive, large-scale, complex enterprises where diverse actors collaborate to deliver an intended outcome (Flyvbjerg, 2014). In this respect, megaprojects offer a rich setting for transitions research to study the interplay of institutions, actors and technology due to their longevity, pervasiveness and embeddedness (Blomquist and Packendorff, 1998; Brookes et al., 2017). Their institutional

complexity is seen in their front-end management, the promoter's role (Gil and Pinto, 2016), their embeddedness (Blomquist and Packendorff, 1998) and involvement of numerous external stakeholders, such as market and government policy, which are also core regime elements.

Moreover, megaprojects exhibit a range of processes and attributes that are conducive to their conceptualisation as niches for transitions research. Government support and the award of contracts to project consortiums constitute a form of protection from established industry structures, dominant technologies and infrastructures, guiding principles and socio-cognitive processes, and markets and dominant user practices. These were potentially factors to be overcome in the case of the megaprojects and thus a form of shielding had to be put in place. Government support and project procurement conditional on digital technology use was instrumental in bringing about change at the industry level. Then, these megaprojects can be conceptualised as niches created around the digital technologies to provide the "shielding" necessary (A. Smith and Raven, 2012) for it to follow a distinct evolutionary trajectory that was subject to certain technology selection criteria. Selection criteria in the niches-megaprojects as will be discussed in Section 4 were different and evolved progressively with each megaproject.¹

The duration of megaprojects and their delivery phases span across years or even decades, and may give rise to interorganizational relations and learning processes similar to those of permanent organizations that participate in niches. Projects are conduits for knowledge creation due to their transience and interdisciplinary nature (Gann and Salter, 2000; Grabher, 2004; Hobday, 2000). Essential learning processes arise at the interface between a project and the organizations, networks, and institutions in which it is embedded (Prencipe and Tell, 2001). Inter-project learning depends on organizational structures between projects, interproject assimilation practices, and actor alignment that facilitates the relationship with other projects (Lundin et al., 2015; Sense and Antoni, 2003). Aggregation dynamics may develop through organizational adoption and adaptation to technology, and the ways partners relate to projects (Manning and Sydow, 2011). The variety of collaborative paths and projects that may develop over time (Brady and Davies, 2004), may contribute to a variety of scale up, aggregation trajectories, and industry wide impact.

The implication is that no project unfolds in an organizational field vacuum, so it needs to be conceptualized as a history-dependant and institutionally-embedded unit of analysis to explain project success or failure better (Biesenthal et al., 2018; Engwall, 2003). To this end, the study analyses how digital technology implementation in megaprojects aggregates and transforms UK construction.

3. Methodology

3.1. Rationale and case selection

We investigate the relation between megaprojects and system level digitalization in UK construction with a case study approach (George and Bennett, 2004; Yin, 1984), as is appropriate for the analysis of context rich phenomena (Eisenhardt, 1989). A single embedded case study design is used here (Yin, 1984) to analyse six intertwined infrastructure projects in UK (Engwall, 2003). The UK is an ideal setting to explore the relation between megaprojects as niches of sociotechnical transitions, given its long history in well-documented megaprojects and the institutionalisation of digital delivery in 2016 that was later followed by other European countries, such as France and Germany in 2020. Similarly, despite the unique character of UK construction, the UK BIM standards (PAS 1102 suite) was later published, as the ISO 19,650 series adopted as national standards by ISO member states. This makes the UK a model case for digitalization and megaprojects that has already inspired and influenced policy in other countries, demonstrating transferability of findings.

The six embedded cases presented were selected through theoretical sampling (Eisenhardt and Graebner, 2007; Pettigrew, 1990) as they are particularly instrumental in industry transformation through digitalization. These 6 cases were selected as maximum variation sampling (or heterogeneous sampling) to capture a wide range of perspectives relating to the phenomenon to gain greater insights by looking at it from all angles and searching for variation in perspectives (Levy, 2008). In chronological order, the six projects are: (1) the Channel Tunnel Rail Link, (2) Heathrow Terminal 5, (3) London Olympics, (4) Crossrail, (5) Thames Tideway and (6) High Speed Two. Projects 1–4 are completed, and projects 5–6 are ongoing. These projects have not been treated yet as part of a single longitudinal study on industry transformation, but have been the subject of significant PM-related research (Brady and Davies, 2014; Davies et al., 2009; Davies and Mackenzie, 2014; Dodgson et al., 2015; Gaunt, 2017; Genus, 1997; Zerjav et al., 2021).

3.2. Data collection and analysis

The cases offer opportunities for complementary and synergistic data gathering and analysis. The embedded cases of completed projects offer the opportunity to study technology aggregation retrospectively in the context of the overall transition pattern while the study of ongoing cases provides a close-up view on project evolution and the mutual adaptation of technology and organizations (Leonard-Barton, 1988; Tyre and Orlikowski, 1994). The embedded case studies build upon data from different sources that work synergistically together to present a coherent story (Yin, 1984). Data spans 36 years from 1985 to 2021, and have been collected through academic literature, government and industry reports, as well as interviews with eight industry experts, policy makers and practitioners which contribute to the external validity of the research (Sarantakos, 2005). More information about the cases and the associated literature on them analysed is shown in Appendix A for brevity.

¹ The correspondence of megaprojects to niches does not necessarily apply in the analysis of other related systems e.g. transport in the case of Crossrail and High Speed 2. The authors would like to thank a reviewer for raising this point.

Table 1

Profiles and identifiers of the interviewees for external validation of the research.

| Identifier | Current role | Professional Background | Area of expertise / relation to infrastructure | Institutional roles held currently or in the past | Interview dates |
|------------|--|---|---|---|--------------------|
| Int-1 | Chief Scientist at the Building Research Establishment (BRE) and Professor at Higher Education | Chartered Engineer and a Fellow of the Royal Academy of Engineering; | Significant policy and industry expertise | Chief Scientific Advisor for the Department of Communities & Local Government. Chair of BuildingSMART UK, Past President of the Institute of Engineering and technology | 30/04/19 |
| Int-2 | BIM Strategy Manager at HS2 | Computer Scientist and Chartered Project Manager | BIM Strategy Manager at High Speed 2 (HS2) | BIM Consultant at Transport for London (TfL) | 08/05/19 |
| Int-3 | Digital Built Britain (BIM) Advisor at a software provider | Design Engineer | Manager of the Crossrail BIM Advancement Academy | Chair of the Asset Data Dictionary in the UK. Member of the UK BIM Task Group | 14/05/19 |
| Int-4 | Research Associate in Construction Law group in Higher Education | Chartered Architect and Chartered Project Manager | Significant policy and industry expertise | Secretary General in the Association for Consulting Architects (ACA), Advisor for Digital Catapult | 30/05/19 |
| Int-5 | Digital Lead in Cities & Development and Aviation at a consulting firm | Chartered Engineer | Manager at the Heathrow Expansion project | Expert Mappers Panel in Infrastructure at Greater London Authority | 30/05/19 |
| Int-6 | Associate Professor in Project Management and Economics | Royal Institution of Chartered Surveyors (RICS) and Association of Chartered Certified Accountants (ACCA) | Senior Cost Engineer for Translink JV at the Channel Tunnel Project | N/A | 29/07/19 |
| Int-7 | Innovation Director, Thames Tideway | Chartered engineer | Digital Innovation at Thames Tideway, Crossrail and London 2012. | Member of i3P; Advisory to Centre for Digital Built Britain; BIM Task Force; Infrastructure Client Group | 03/03/20 |
| Int-8 | Research and teaching. Higher Education | Architect and academic | Research - Avanti | Leader of Research & Innovation at major industry body. | 30/04/21 |

To complete the case study data collection, eight interviews were conducted with experts on different cases. This sample was kept small in order to avoid impression management from the retrospective cases (Eisenhardt and Graebner, 2007), but also have enough information to triangulate the rest of the analysis. Interviewees have diverse backgrounds and roles (see Table 1): some have direct involvement in past and ongoing infrastructure projects (Int-2,3,5,6,7), three interviewees (Int-1,4,8) have significant industry and policy experience, and two hold key management roles in ongoing infrastructure projects (Int-2,7). Collectively, all of them have deep involvement in all six cases. The high-quality of the interviewees is shown in their boundary-spanning roles, bridging industry, policy and research that constituted them unique experts on the phenomenon. The coding process followed a combination of deductive coding, to identify linkages to the six megaprojects, and inductive coding to establish the role of institutions, actors and digital technologies (Saldana, 2009).

The cases are explored through a process lens (Langley, 1999; Langley et al., 2013). We first describe the institutional setting for the embedded cases. Then, the managerial choices made and emergent processes in each project are described following guidelines for qualitative enquiry (Denzin and Lincoln, 2017). The longitudinal view on the cases reveals the inter-organizational processes at work and the bidirectional influence of project organization and industry level developments. We use a combination of narrative and visual mapping as the strengths of each method counter the weaknesses of the other (Jick, 1979; Johnson et al., 2017). Diagrams do not constitute theory development on their own, but are part of theory building blocks (Griffin, 1993; Sutton and Staw, 1995), and theorization (Langley and Ravasi, 2019; Swedberg, 2017, 2016; Weick, 1995).

4. Findings

4.1. Case institutional setting

The UK construction industry is a regime that has been historically plagued by chronic fragmentation in the coordination and communication of designs and delivery of buildings (Egan, 1998; Latham, 1994), that had resulted in the well-documented performance gap between the design and actual building performance (De Wilde, 2014; Menezes et al., 2012). Such coordination problems are magnified in megaprojects, as each one has its own internal economy, governance structure, and production system (Davies et al., 2009). The consensus view in the industry developed over the past decades is that these challenges have to be addressed (Armitt, 2013; Egan, 1998; Farmer, 2016; Latham, 1994; Sergeeva and Winch, 2020).

In response to this, the 2011 Government Construction Strategy (GCS) focused on the need for widespread adoption of digital technologies such as Building Information modelling (BIM) (Cabinet Office, 2011). Through this milestone development, the Government mandated that BIM Level 2 should be used on all public sector projects starting after April 2016 (GCCG, 2011). This mandate has significantly affected the project delivery of publicly-procured assets such as transportation (Whyte, 2019). It calls for all project

and asset information, documentation and data to be digital and specifies the collaborative use of BIM models and digital objects (Papadonikolaki et al. 2019). To support these efforts, the UK government created the UK BIM Task Group which was publicly-funded until 2017. Thereafter a period of intensive standardization followed, with high levels of information exchange, deliberations, and strong leadership. In particular, the British Standards Institute (BSI) issued a suite of Publicly Available Specifications (PAS) number 1192 with parts 2–6 published between 2013 and 2015 (that later became ISO).

In 2013, the Government issued the 'Construction 2025: Industry Strategy' to reaffirm its strong support for BIM. It emphasised the joint commitment and close collaboration of government and industry to the BIM vision and programme. The strategy outlined a vision that all government procured projects would be delivered through a digitally-enabled delivery and lead eventually to a wider offsite manufacturing strategy. In 2016, the Cabinet Office and the Infrastructure and Projects Authority (IPA) issued the 2016–2020 GSC which built upon the 2011 strategy, emphasising BIM and Digital Construction as "an important part of the strategy and is helping to increase productivity and collaboration through technology" (Office, 2016). Presently, BIM is considered the most representative digital technology and information aggregator in construction globally. The Farmer Review (2016) commissioned by the Construction Leadership Council at the request of the UK Government resonated with earlier reports by Wolstenholme et al. (2009), Egan (1998) and Latham (1994) attributing productivity losses and lack of collaboration in construction to a paucity of innovation and widespread skills shortage. The review called for action in light of the newly-announced Thames Tideway and High Speed 2 projects in London.

4.2. Channel tunnel (CT) (1985–1994)

At the time of construction in the mid- '80 s of the first megaproject studied here, the Channel Tunnel, the UK construction industry was characterised by inefficiency and low digitisation. The discourse on improvements in UK construction included visions of partnering, supply chain management, and lean manufacturing philosophy, all of which were imported from other sectors, such as manufacturing (Bresnen and Marshall, 2001; Pavitt, 1984; Reichstein et al., 2005). In this context, CT was seen as a key infrastructure project for change in UK construction sector. A consortium comprising Bechtel, Arup, Systra, and Halcrow was established to run the \$10.3billion Channel Tunnel Rail Link project linking UK to Europe's high-speed rail network (Davies et al., 2009). The technical complexity of CT pushed the boundaries of technology innovation in UK construction.

Back then, the method to produce and share design information was through 2D plans (Harty and Whyte, 2010). Digital innovations introduced from the US were a radical departure from prior established ways of working (Genus, 1997). They enabled the exchange of building information through Computer-Aided Design (CAD) applications, the creation of a virtual building through software before commencement of on-site work (Harty, 2005), and limited error-prone human interventions (Eastman, 1999). As Int-6 explained, 3D modelling tools (Pöttler, 1992) and other computer-based tools to deal with interoperability issues amongst different systems were tested in CT for project planning, cost management, procurement systems, and for facility management after construction.

As Int-6 explained, the implementation of 3D CAD in drafting departments required broader organizational adaptations and support, as its repercussions reverberated throughout the inter-organizational network of project actors. Instead of 2D plans to be used and adjusted on-site, 3D CAD required full design specification upfront, which left no room for design alterations during construction. Innovations used in CT paved the way for two subsequent CT phases in 2003 and 2007 (Arup, 2004; Pollalis and Georgoulias, 2008). The CT project informed the Latham (1994) and Egan (1998) reports that called for increased supply chain integration and collaboration and digital technology use to improve industry performance (Papadonikolaki and Wamelink, 2017).

4.3. Heathrow terminal 5 (T5) (1999-2008)

T5 was a \$8.5 billion project for a new terminal at Heathrow airport to increase its annual capacity from 67 million to 95 million passengers. Due to its size, the T5 project became a program of industry-wide change in the UK (Davies et al., 2009). British Airports Authority (BAA) committed to 3D CAD implementation for coordination of design and construction, and information management across the whole project (Harty, 2005). BAA in consultation with partners pursued an integrated team approach to technology introduction and use in T5, that substituted some traditional drafting practices. This was shaped by T5 project managers who drew on the previous Heathrow T4 project (Harty, 2005). Personnel transfer from CT to T5 facilitated the further "evolution of digital technologies there" (Int-6). This approach was enabled by inter-compatible design and drafting software packages, mediated by a document management database.

The use of BIM in T5 aimed to create a single-model environment to improve coordination in the construction process, systems integration and information flows across actors, so that problems could be identified prior to work on-site where errors are costly (Davies et al., 2009). In T5, BIM can be described as a systemic innovation, as its consequences spread across multiple partner tiers (Harty, 2005). BIM is not a singular, stand-alone technology like CAD, but a more platform-like and thus disruptive innovation at the organizational level (Morgan, 2019). It is the result of evolving efforts by industry consortia, such as BuildingSMART to standardize building information (East and Smith, 2016).

The benefits of BIM were enough to convince project engineers and drafters they had to change their practices but were not enough to drive technology implementation at the organizational level. Considerable work was necessary to organise a coherent and inclusive system of technologies and practices in alignment to the diverse expectations and visions of the project (Harty, 2005). Negotiations between project actors indicated the software packages as necessary to align the BAA vision and the visions of services engineers and drafters. However, existing software packages could not facilitate the BAA vision and meet the selection criteria of project actors. This required that software had to transform from being primarily a design tool for 3D design to a manufacturing and control tool. Such

digital innovations were highly influential in allowing the design to be engineered: for instance, model buckling analysis and sophisticated wind analysis were made possible on the complex roof structure of the terminal (Arup, 2006).

T5 project activities shaped software package functionality and triggered a cycle of mutual technology and adaptation (Morgan, 2019). Technologies were incorporated into practices that already utilized a diverse network of material and digital artifacts. Moving towards a digitally orientated process was not just a simple case of technology substitution but fundamentally changing the practices they constituted (Harty, 2005; Harty and Whyte, 2010). However, configurations of practices already in place in T5 proved robust so the more initiatives to change challenged established practices and ways of work, the more they were resisted (Harty and Whyte, 2010). Eventually, the digital link between design and fabrication had to be dropped due to such difficulties.

4.4. London olympics 2012 (2005-2012)

London Olympics transformed Stratford in east London, into a 2.5 km² Olympic Park with an athletics stadium, aquatics centre and velodrome. The London Olympics project (LO) drew on the lessons of T5 with digital technologies and software tools used for coordination at each phase of system integration project design and documentation management, e.g. in ProjectWise (Davies et al., 2009; Davies and Mackenzie, 2014). At the same time, systems integration related to collaborative teamwork and managing uncertainty by mutual adjustment and collection of new information. Legacy issues were widely considered in LO, described as a 'field-configuring event' (Thiel and Grabher, 2015). Several senior managers from London Olympics transferred their experience and innovative ideas on digital innovation to Crossrail, including Andrew Wolstenholme chief executive at Crossrail and former programme director at T5 (Wolstenholme et al., 2009).

Simultaneously, the UK government took steps to promote digitalization of UK construction through institutional projects (Holm, 1995). This included Avanti, a research project whose aim was to enable effective team collaboration through technology (Morgan, 2017). It was an inter-institutional collaboration amongst the Department of Trade and Industry supported by large UK construction firms, universities and R&D organizations, and BuildingSMART (formerly International Alliance for Interoperability). Funded by the UK government, Avanti drew several major industry partners together to create a culture, processes and digital tools to enable team collaboration (Morgan, 2017) through the use of two-dimensional (2D) digital design.

Avanti became the basis of the BIM British Standard BS1192. Such was the industry enthusiasm for Avanti, that it continued to be supported after government funding had run out. The Avanti principles were subsequently used in the Victoria Station Upgrade project to manage information flows and partner collaboration and "get *people* to trust that data" (Int-3). The significance of Avanti lay not only in its influence in establishing data standards but also in addressing cultural change: "in an industry where adversarial practices ruled...we paid a lot of attention to establishing a collaborative culture in Avanti, with hindsight this was as great an achievement of the project as some of the more technical advances". (Int-8).

4.5. Crossrail (2008–2022)

Crossrail is the biggest civil engineering project in Europe and the basis for several digital innovations in construction (Arup, 2012). Crossrail will provide high-frequency suburban passenger service crossing London from west to east. Crossrail started with 2D deliverables before the 2011 UK BIM mandate, but was completed using BIM and 3D digital deliverables. It was amongst the first UK projects to become PAS1192-compliant and use BIM as a digital platform for other innovations, such as laser scanning and augmented reality. However, the BIM Level 2 criteria were seen as obscure and too "*open to interpretation, to be legally enforceable, but we were all aiming towards that 2016 deadline*" resulting in standards not "*as clear and concise as we would want*" (Int-3), which created challenges in the project and technology uptake. The innovation strategy followed at Crossrail received considerable scholarly and practitioner attention (DeBarro et al., 2015). The Innovate 18 programme, was set up in association with Imperial College as a formal element of Crossrail delivery aiming to create a collaborative innovation culture across the project, partners and suppliers (DeBarro et al., 2015; Pelton et al., 2017).

The leadership shown by the government was catalytic to reconfigure work with digital innovations and "really changed where we were." (Int-3). From the perspective of the private sector, the government both local and national did some brilliant work to modernise planning policy and drive "digital transformation in large-scale infrastructure through the BIM policy" (Int-5). To further digital innovation in the industry, a BIM Academy was established in partnership with Bentley software to support the Government Construction Strategy, increase the use of BIM and create a lasting legacy of best practices in the industry (Munsi, 2012). The Academy training helps industry constructors to transfer their knowledge to other major projects such as HS2 and TT. Both LO and Crossrail had a strong emphasis on passing on lessons learnt to later infrastructure projects. This is particularly evident in HS2 and Thames Tideway (Pelton et al., 2017). Many senior managers from the London Olympics and Crossrail went on to work at TT or HS2, thereby transferring knowledge on digital technologies between these megaprojects.

4.6. Thames tideway (TT) (2012-2023)

The TT project is a tunnel running mostly under the tidal section of the River Thames through central London to capture, store and convey almost all the raw sewage and rainwater that currently overflows into the river. The TT is the first megaproject to adhere to BIM Level 2 standards. It is a landmark development in digital modelling, requiring that all project actors use digital model-based delivery (Gaunt, 2017). Significant efforts were made during initial procurement to assess the risks implicit in BIM delivery. However, while contractors were required to use BIM delivery, the TT team chose to adopt a 'technology agnostic' approach during tendering. The



Fig. 1. Timeline of digital technologies in UK construction influencing and being influenced by institutions, actors and megaprojects as sociotechnical systems (dashed lines indicate authorisation dates).

reason was prior experience from projects including Crossrail "where they had imposed a particular bespoke software or the client hosted everything' meaning that 'the risk was with the client." (Int-7). The TT team defined project deliverables from a digital blueprint and the BIM Execution Plan but encouraged contractors to use tools they already had expertise in so that they would "be able to go back to their own home organisations and utilise the latest, greatest software, where possible." (Int-7).

The TT team is also heavily involved in the Infrastructure Client Group to promote digitalization in construction and align "to the wider UK rollout, especially infrastructure" (Int-7). The aim is to draw on the lessons of the Crossrail and Olympics projects, create a digital learning legacy for construction, pass on lessons to HS2 and share best practices. 2016 marked a turning point with less government involvement but industry sentiment was that government support was still necessary "to try and drive that leadership from the top" (Int-3). Initially, there were great expectations that the industry would follow through with best practices in BIM Level 2 and the insurance sector would play a key part, but it became apparent that time was necessary "to see that properly emerge. I believe that's starting to happen, however." (Int-1).

4.7. High speed two (HS2) (Phase 1: authorised in 2017 2029-2033)

The first phase of HS2 is a major high-speed rail line that links London and Birmingham. Building upon developments from previous projects, but also benefitting by movement of experienced staff from LO and Crossrail, HS2 will collect and utilise knowledge to construct the digital twin of the asset, that mirrors the physical infrastructure and enables scenario simulations. This is one of several industry initiatives in Digital Built Britain, that "*can certainly go a long way, and that would probably be HS2's digital legacy.*" (Int-2). In this regard the HS2 team works closely with government departments, with major infrastructure programmes, and professional institutions to try and "*be proactive in developing standards and making sure that our requirements, our expectations from our supply chain, are reasonable.*" (Int-2).

The HS2 project has redefined digital innovation in UK construction by "looking at developing competencies which are not just limited to BIM in a sense that it's been defined as just a modelling or a CAD tool." (Int-2). There is a growing awareness of the digital legacy the project creates, and it becomes clear that it is necessary to "look at this as a wider data management piece" (Int-2). In this respect, standards like ISO19650 are a good starting point because they look at "at information management in its wider sense and its wider context" (Int-2). They are also pioneering solutions that benefit the entire industry and aim to create lasting change such as the HS2 BIM upskilling platform (HS2, 2019).

Although regulation activities stopped with the 2016 mandate, organizations have since then increasingly committed to digital practices. The perception in the industry changed from "Digital is something interesting. It's something on the side," to "It's core to our business ... It's not a case of digital is a nice thing you do in your spare time." (Int-5). This indicates the increased legitimacy of digital competences and practices. Despite the marked change, the five years from 2011 to 2016 "was an incredibly short amount of time to change an industry which, [...] I think that hasn't entirely happened yet. I think we still need a lot more leadership. We need a lot more guidance." (Int-3). Indicative of the progress made is the refinement of ISO19650 which still is "not as enforceable in the client world as I think it should be." (Int-3).

Table 2

Transition pathways of digital technologies through UK megaprojects along three categories (institutions, actors and technologies), based on framework by Geels et al. (2016).

| Transition pathways | Rules and institutions | Actors and social groups | Digital technologies and STS |
|--------------------------------|---|--|--|
| Reconfiguration (-1998) | Limited institutional change from government- sponsored industry reports calling for change No rules developed | Firms become sporadically aware of the need for change No new entrants | In complex infrastructure projects, the need for technological change is accentuated Digital tools complement and substitute existing |
| Transformation (1998–2011) | Institutional change by sponsoring an institutional and collaborative project to inspire change Project outputs become basis of British | Incumbents reorient incrementally by adjusting their processes New government-sponsored groups and communities to drive change emerge | Incremental progress in existing technologies Incorporation of symbiotic niche- innovations and add-ons that change the |
| Substitution (2011-present) | Standards Substantial institutional involvement via mandates for new technology in public procurement Issuing of suites of standards to lead change | Incumbent firms reorient substantially, to new technology and business model New alliances between incumbents and new entrants | processes From initial add-ons to new hybrids of new and existing technologies Partial or full technical substitution of tools that brings new processes |

Despite the UK government leadership, the "local government doesn't seem to actually drive the mandate, that it was pushed through for 2016. So, we can already see setbacks. We want it to work, but we have really setbacks that we need to overcome" (Int-4). Industry digitalization requires collective action "*So, those early adopters in terms of procurement and use of BIM used - to the majority, again, to stage open book tender. So, technology could actually assist but, again, it needs the combination of that collaborative work that we need to bring together." (Int-4). For example, in some of the pilot projects in the Ministry of Justice "they actually brought in the supply chain, and even the operators, and they were involved in the dialogue very early on in the design stage to be able to really." (Int-4). Further evidence of industry change can be found in 'Project 13', a recent industry-led collaborative endeavour that seeks to develop a new business model, based on an enterprise, not on traditional transactional arrangements, to improve UK construction. amongst the ambitions of the Infrastructure Client Group (ICG) is to "establish standards that all ICG members will sign up to and enforce across all of their projects".*

5. Discussion

5.1. Aggregation of digital technologies

To address the first question, which focuses on system-wide transformational impacts in UK construction, the analysis indicates that the evolution of digital technologies and related standards in UK construction has gone through several stages, punctuated by regulation, and marked by the move of scholars and practitioners towards a broader scope for BIM (Dainty et al., 2017; Morgan, 2017). Drawing on the preceding sections, Fig. 1 summarises the process of aggregation of digital transition in UK construction and juxtaposes (1) rules and institutions, (2) actors and social groups and (3) digital technologies with the temporal sequence of megaprojects, and policy reports. By viewing infrastructure projects as niches, a key contribution of this study is apparent. Our analysis suggests that the transformation process ripples outwards from the regime core to its periphery. This is interesting but in addition one could argue that this transition does not adhere closely to one of the proposed stylised pathways (Geels and Schot, 2007). Rather, we distinguish three phases (Fig. 1) in this process for which a tentative case can be made as to its correspondence to a pathway.

In the first phase, digital technologies are adopted to enhance design and communication processes in construction supply chains so that the production and delivery onsite of complete blueprints is done with minimum errors. The incremental adoption of technologies in existing design and construction incumbent practices resembles the *reconfiguration* pathway where innovations are adopted in a symbiotic way in a regime to solve local problems, in our case the challenges that actors in individual megaprojects face. This process triggers subsequently further adjustments in the basic architecture of the regime. The second phase has elements of *transformation* pathway, as innovations are still being developed and refined and at the same time regime actors modify the development direction of their capabilities to align them with new standards they should adhere to. This was in part because technologies where not endemic to construction. Section 4 documents how industry incumbents engaged with software vendors in the development of *substitution* pathway in that BIM solutions are available to industry actors and are developed sufficiently and specifically to their needs. To the extent that actors commit the necessary resources, BIM can potentially substitute completely previous practices and processes industry that actors have in place in their construction supply chains. Table 2 shows the sociotechnical transitions in digital technologies through UK megaprojects.

5.2. Megaprojects in sociotechnical transitions processes

To address the second question on how technologies in megaprojects drive sociotechnical transitions and vice versa, we start from framing infrastructure megaprojects as niches where incumbent actors engage in joint endeavours, standardization, and cooperation with the government. The scale of megaprojects magnifies coordination and performance problems due to industry fragmentation and intensifies partner tensions. Technology adoption triggers the digitalization process and the six megaprojects drive the aggregation

process. They provide the impetus for digital technology implementation, a test bed for the refinement of its selection criteria, and the development of new regulations and standards.

Sociotechnical changes at the firm and regime levels influence mutually constitutive relations between institutions, organizations, projects and actors (Morgan, 2019; Shibeika and Harty, 2015). The flow of key personnel across projects and institutions and their boundary-spanning capabilities drives and is driven by digital innovation and standardization (Koskinen, 2008; Levina and Vaast, 2005). Standardization is punctuated by two milestones, the Egan Report (1998) and the 2011 mandate for digital project delivery. Particularly after 2011, the Crossrail, TT and HS2 consortia formed strategic alliances with the UK government to drive digital innovation and standardization which were essential for digital technology evolution.

To understand how the digitalization process begun at the regime core and rippled out to the periphery of the construction industry, one has to understand that large incumbent firms in UK construction were in a position to impose their project related requirements to their 1st and 2nd tier suppliers. These requirements emanated from government strategic procurement policies, institutional developments and standards, for example, the Avanti project (2001–2005) set the foundation for PAS1192 standards issued during 2013–2015. The regime/organizational field core to periphery dynamics then unfold over time because BIM adoption and implementation at the project level require collaboration between partner firms (Harty, 2005) and trigger organizational changes (Leonard-Barton, 1988; Morgan, 2019; Peansupap and Walker, 2007; Tyre and Orlikowski, 1994). The establishment of local standards of practice and organizational changes that initially span design to actual construction and post commission work, eventually aggregate to institutional changes that ripple across the industry (Boland et al., 2007). They have system wide impact as they change established organizational practices and technology selection criteria, and stimulate the development and use of sector specific software packages that are considered the norm today.

5.3. Theoretical implications

The case offers a clear example of how incumbents in capital-intensive industries survive the challenge an innovation may present (Ansari and Krop, 2012; Bergek et al., 2013; Berggren et al., 2015; Cohen and Tripsas, 2018; Hill and Rothaermel, 2003). This would suggest that incumbents' chances of survival during a transition are better than is implied in most transitions research. Indeed, industrial organization research indicates industry incumbents have better survival chances than entrants of an industry in its mature phase (Agarwal et al., 2002; Agarwal and Gort, 1996; Christensen et al., 1998; Klepper, 1997, 1996; Malerba and Orsenigo, 1996; Suarez and Utterback, 1995). The importance of this is significant in terms of transitions research as large incumbent organizations often have a large societal impact (Matten and Crane, 2005), that can stall or accelerate a transition.

The case of megaprojects deviates in a number of ways from a narrative where new entrants and outsiders challenge dominant incumbent actors and regimes. Megaprojects are not peripheral outsiders but instead protected innovative niches at the centre of the organizational field and their effect ripples outward towards its periphery. This provides fertile ground for transitions research that may uncover a greater variety in aggregation patterns and transition pathways (A. Smith and Raven, 2012). First, the positive role of incumbents that catalyse rather than raise barriers to change indicates that scholars should analyse symmetrically niche-to-regime activities and regime-to-niche activities. This may include the strategic reorientation of incumbent actors in the focal regime or capital-intensive industries.

Second, the technology implementation and use in megaprojects is in stark contrast to the system innovation pattern that begins with diffuse niche visions, open-ended experimentation, expectations and diverse search and technology development processes and develop to present a credible alternative to the dominant regime (Kemp et al., 1998; J. Schot and Geels, 2008; A. Smith and Raven, 2012; Turnheim and Geels, 2019). This may be a feature pertinent to infrastructure projects that need prior consensus and agreement to start, where failure is simply not an option, in part because of their high profile. Moreover, the digitalization process seems to become broad and divergent rather than become more specific and convergent in scope, as suggested in the literature (J. Schot and Geels, 2008). Digital technologies become linked to multiple possibilities e.g. advanced applications of BIM, additive manufacturing, artificial intelligence and robotics, the internet of things, big data analytics and blockchain technology.

Finally, bi-directional exchange between PM research and transition research offers some promising avenues through structuration theory (Bresnen et al., 2005; Floricel et al., 2014; Manning, 2008; Windeler and Sydow, 2001), and multi-level theorization that includes the firm level and the institutional - organizational field level (Gann and Salter, 2000; Lundin et al., 2015; Sydow and Braun, 2018; Winch, 1998), to examine temporalities and temporary/permanent organizing interaction at macro-meso-micro levels, the relation of projects to institutions (Biesenthal et al., 2018; Qiu et al., 2019; Söderlund and Sydow, 2019; Winch and Maytor-ena-Sanchez, 2020), and project governance (Ahola et al., 2014; Müller et al., 2016; Too and Weaver, 2014). With macro scale initiatives for macro scale changes as a starting point, researchers in transitions research and PM could look for other similar theoretical tensions and use them to stimulate better theorizing (Swedberg, 2017; Weick, 1995), and the development of more encompassing theories (Poole and van de Ven, 1989).

5.4. Implications for policy and practice

Our cases show the role of strong guidance, consensus, early direction-setting, and sustained resource commitments, e.g. national R&D funding, national strategic procurement for transition processes. The industry reports exemplify the strategic consensus and the vision for the industry (Egan, 1998; Farmer, 2016; Latham, 1994; Wolstenholme et al., 2009). They provide a strong impetus for change supported by government level standardisation initiatives such as the Avanti project, that is catalysed by the six megaprojects. These projects serve as milestones because of the significance of the technology verification and exemplification they enable, but also

because their design implications were systematically replicated in later projects, supporting stepwise changes in the overall system trajectory. These megaprojects together with policy on digital delivery of projects (announced in 2011 and enforced in 2016) catalysed the speed of technology diffusion and organizational change and its directionality. To this end policy acted as the protective shielding of the megaproject-niches.

Across the 30-year longitudinal process of industry digital evolution studied here, we see the rich interplay of government and other policy bodies adopting role of 'client' or funder for the megaprojects and that of policy producer (for example through the Latham and Egan reports and BIM Mandate). Our data suggests that it is this interplay of roles that guides the evolution of industry practice, and is guided by it (Morgan, 2019). The recognition of these roles and the mutually constitutive relationship between them is significant in developing effective policy interventions that encourage evolutionary processes.

The purposeful character and focus of industry change efforts indicates that each megaproject can be considered as a niche for analysis purposes (Turnheim and Geels, 2019). Megaprojects with the extensive range of organisations and institutions involved, can function as institutional change levers or even political projects by invoking the collective action of political actors (Holm, 1995). For example, megaproject executives were aware of and keen to leave a 'digital legacy' (Munsi, 2012). It would seem that the proposition that single projects are not so important for niche development does not apply in our case (Hoogma et al., 2002).

While the sheer scale of megaprojects may seemingly provide little scope to consider agency, managerial cognition in fact plays a key role in identifying capabilities suitable for the environment where an organization operates (Lavie, 2006). The case of Wolstenholme is illustrative in this respect. Managerial cognition enables organizations to reconfigure their capabilities through substitution, evolution or transformation (Cohen and Tripsas, 2018; Lavie, 2006). A leader's interpretations of the external landscape will affect how they see their own organization's abilities to respond to it (Milliken, 1987). The cases highlight that the match between organizational capabilities and a market opportunity is not sufficient for change if leadership beliefs e.g. of Wolstenholme, are not aligned with the opportunity (Eggers and Kaplan, 2013, 2009; Tripsas and Gavetti, 2000).

5.5. Future research avenues

The sociotechnical perspective and analysis of projects as part of broader system level changes may be of interest to PM scholars that view projects primarily as distinct units of analysis, and vice versa transition scholars may stand to benefit from the vast PM literature. The potential for bidirectional cross overs is considerable. For example, the subject of interorganizational relations and actor alignment is important from a PM perspective but underdeveloped (Sydow and Braun, 2018). In this regard, PM research can benefit from a socio-technical perspective (Biesenthal et al., 2018), in order to highlight the role of organizational inertia and path dependency, and broader social, economic and historical processes that lie beyond the project management scope.

Research can extend beyond the study of individual projects to a project lineage, knowledge transfer from one project to another, to permanent organizations and the institutional context in which they are embedded (Engwall, 2003; Windeler and Sydow, 2001). The analysis of inter-project knowledge transfer could benefit from transitions research on how knowledge about project coordination practices accumulates, aggregates and changes the organizational field, and, in turn, it shapes these practices (Geels and Deuten, 2006; J. Schot and Geels, 2008; A. Smith and Raven, 2012). Similarly, transitions research can draw on the vast PM literature on factors that inhibit knowledge development and transfer between niche projects (DeFillippi and Arthur, 1998; Gann and Salter, 2000; Keegan and Turner, 2002, 2001; Koskinen et al., 2003; Williams, 2008).

Moreover, another bidirectional avenue between transitions research and PM research lies at the institutional level of analysis (Fuenfschilling and Truffer, 2014). PM scholars advocate for more fine-grained analyses of projects in their institutional context (Dille and Söderlund, 2011; Engwall, 2003; Grabher, 2004; Morris and Geraldi, 2011). The successful completion of projects and megaprojects in particular, requires considerable institutional entrepreneurship and strong project ownership or sponsorship (Biesenthal et al., 2018; Brady and Davies, 2014). This raises also the issue of megaproject timing in relation to a favourable or unfavourable institutional context (Dille and Söderlund, 2011), and is resonant with recent calls to explore the role of large-scale technological options for low carbon transitions (Sovacool and Geels, 2021; Turnheim and Geels, 2019), and the importance of windows of opportunity, timing and sequence for transitions and projects.

In summary, bidirectional exchange between PM research and transition research offers some promising avenues through structuration theory (Bresnen et al., 2005; Floricel et al., 2014; Windeler and Sydow, 2001), and multi-level theorization that includes the firm level and the institutional - organizational field level (Lundin et al., 2015; Sydow and Braun, 2018; Winch, 1998), to examine temporalities and temporary/permanent organizing interaction at macro-meso-micro levels, the relation of projects to institutions (Söderlund and Sydow, 2019), and project governance (Ahola et al., 2014). This opens a wider vista of research on the interface between PM and transitions research on megaprojects and a contextual understanding of how they are situated in a large project-based context.

6. Conclusions

The framing of digital technology adoption in UK construction as a sociotechnical transition process with megaprojects framed as niches, offers an interesting alternative to the dominant narrative in transition research where sociotechnical processes at the periphery of an organizational field, accumulate momentum, scale up and eventually transform it. In contrast to this dominant narrative, the case analysed in this paper indicates that infrastructure systems may exhibit the reverse process: from core to periphery. The analysis of UK megaprojects shows how digital technologies introduced in construction, shape and are shaped by the interplay of institutional and organizational change processes of incumbents at the regime core. They are shown to enable and drive, rather than

Table 3

The six sub-units of the embedded case study: UK construction infrastructure projects.²¹

| Major UK infrastructure projects | Project dates | Type of project | Digital innovation | Literature sources (excluding interviews) |
|--|------------------|---------------------|---|--|
| Channel Tunnel (CT) | 1985–1994 | Rail infrastructure | Separate modelling solutions for 3D design, cost | Pöttler (1992); Genus (1997); Arup (2004); |
| | | | management and procurement | Pollalis and Georgoulias (2008). |
| Heathrow Terminal 5 | 1999-2008 | Airport building | Efforts to consolidate digital tools such as CAD in | Davies et al. (2009); Harty and Whyte and |
| (T5) | | | an online 'single-model environment' and | Lobo (2010); Arup (2006); Davies et al. |
| | | | recognition of 'hybrid practices' | (2016); Gil et al. (2012). |
| London Olympics | 2005-2012 | Sports facilities | Application of developments from CT and T5 on | Davies and Mackenzie (2014); Davies et al. |
| 2012 | | | technology support and knowledge management | (2014); Thiel and Grabher (2015). |
| | | | about digital innovations | |
| Crossrail | 2008-2021 | Rail infrastructure | Integration of digital tools such as Building | DeBarro et al. (2015); Dodgson et al. |
| | | | Information Modelling (BIM) through online | (2015); Arup (2012); Pelton et al. (2017). |
| | | | Common Data Environments (CDE) | |
| Thames Tideway | 2012-2023 | Water disposal | BIM as contracted deliverable. Team will hand | Gaunt (2017); Morgan (2020); Tideway |
| | | infrastructure | over a digital copy of the infrastructure asset, | (2021). |
| | | | along with the physical version | |
| High Speed Two (or | 2017-2026 | Rail infrastructure | Integration of BIM with virtual reality aiming to | Podkolinski (2021); HS2 (2019, 2021). |
| HS2) | | | develop a digital twin of the rail | |

inhibit the aggregation trajectory of project based digital innovation. The trajectory is underlined by actors whose mobility across megaprojects and institutions is instrumental in the development of digital technology standards and regulation.

The sociotechnical perspective and analysis of projects as part of broader system level changes may be of interest to project management scholars that view projects primarily as distinct units of analysis. The implication of this study is to highlight potential conceptual cross overs between PM and transition research. Understanding the inter-relationships amongst key infrastructure projects, institutions, actors and how they influence digital innovation is a timely subject for both project management and transition research. It will help prepare for and identify patterns and opportunities to manage the unprecedented pace of emergent digital technologies that influence the industry.

These findings are also valuable for other sectors. The built environment allows us to study this relatively slow transformation over three decades and identify mechanisms and inter-relations that are hardly noticeable in other sectors, where the pace of innovation is more accelerated. In this respect this single embedded case may be shed light on contemporaneous projectification processes in other systems. Digitalization has been seen as a catalyst and an important means to address Grand Challenges such as Climate Change and Clean Growth. This study offers transferable findings for energy and pharmaceuticals disciplines that are also highly-regulated with a strong regime element that is challenged through digitalization and policy acting as protective shields of niches.

Declaration of Competing Interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Data availability

The data that has been used is confidential.

Acknowledgements

The author(s) disclosed receipt of the following financial support for the research, authorship, and/or publication of this article: This work was supported by the Engineering and Physical Sciences Research Council (EPSRC) (grant number EP/N509577/1, 178701).

The author(s) would like to thank Fernando Suarez and Geert Verbong for comments on early versions of the paper.

Appendix A

Table 3

References

Adner, R., Levinthal, D., 2001. Demand heterogeneity and technology evolution: implication for product and process innovation. Manage. Sci. 47, 611–628.

Environmental Innovation and Societal Transitions 48 (2023) 100728

Agarwal, R., Gort, M., 1996. The evolution of markets and entry, exit and survival of firms. Rev. Econ. Stat. 78, 489-498.

Agarwal, R., Sarkar, M.B., Echambadi, R., 2002. The conditioning effect on firm survival: an industry life cycle approach. Acad. Manag. J. 45, 971–994.

Ahola, T., Ruuska, I., Artto, K., Kujala, J., 2014. What is project governance and what are its origins? Int. J. Project Manage. 32, 1321-1332.

Ansari, S., Krop, P., 2012. Incumbent performance in the face of a radical innovation: towards a framework for incumbent challenger dynamics. Res. Policy 41, 1357-1374.

Armitt, J., 2013. The Armitt Review. Labour's Policy Review.

Arup, 2004. Channel tunnel rail link special link. Arup J. London, UK.

Arup, 2006. Arup 60th anniversary year. Arup J. London, UK.

Arup, 2012. The London special issue. Arup J. London, UK.

Astley, W.G., 1985. The two ecologies: population and community perspectives on organizational evolution. Adm. Sci. Q. 30, 224-241.

Bergek, A., Berggren, C., Magnusson, T., Hobday, M., 2013. Technological discontinuities and the challenge for incumbent firms: destruction, disruption or creative accumulation? Res. Policy 42, 1210-1224.

Berggren, C., Magnusson, T., Sushandoyo, D., 2015. Transition pathways revisited: established firms as multi-level actors in the heavy vehicle industry. Res. Policy 44, 1017-1028

Biesenthal, C., Clegg, S., Mahalingam, A., Sankaran, S., 2018. Applying institutional theories to managing megaprojects. Int. J. Project Manage. 36, 43–54.

Blomquist, T., Packendorff, J., 1998. Learning from renewal projects: content, context and embeddedness. In: Lundin, R.A., Midler, C. (Eds.), Projects As Arenas for Renewal and Learning Processes. Springer US, Boston, MA, pp. 37-46. https://doi.org/10.1007/978-1-4615-5691-6_4.

Bogers, M., Afuah, A., Bastian, B., 2010. Users as innovators: a review, critique, and future research directions. J. Manage 36, 857-875.

Boland, R.J., Lyytinen, K., Yoo, Y., 2007. Wakes of innovation in project networks: the case of digital 3-D representations in architecture, engineering, and construction. Organ. Sci. 18, 631-647.

Brady, T., Davies, A., 2004. Building project capabilities: from exploratory to exploitative learning. Organ. Stud. 25 (9), 1601–1621.

Brady, T., Davies, A., 2014. Managing structural and dynamic complexity: a tale of two projects. Project Manag. J. 45, 21-38.

Bresnen, M., Goussevskaia, A., Swan, J., 2005. Implementing change in construction project organizations: exploring the interplay between structure and agency. Build. Res. Inf. 33, 547-560.

Bresnen, M., Marshall, N., 2001. Understanding the diffusion and application of new management ideas in construction. Eng. Construct. Architect. Manag. 8, 335-345.

Brookes, N., Sage, D., Dainty, A., Locatelli, G., Whyte, J., 2017. An island of constancy in a sea of change: rethinking project temporalities with long-term megaprojects. Int. J. Project Manage. 35, 1213-1224.

Budde, B., Konrad, K., 2019. Tentative governing of fuel cell innovation in a dynamic network of expectations. Res. Policy 48, 1098–1112. https://doi.org/10.1016/j. respol.2019.01.007.

Cabinet Office, 2011. Government construction strategy. London: Cabinet Office. Available from: https://assets.publishing.service.gov.uk/government/uploads/ system/uploads/attachment_data/file/61152/Government-Construction-Strategy 0.pdf.

Chasen, S.H., 1981. Historical highlights of interactive computer graphics. Mech. Eng. 103, 32-41.

Christensen, C.M., Suarez, F.F., Utterback, J.M., 1998. Strategies for survival in fast-changing industries. Manage. Sci. 44, S207–S220.

Ciarli, T., Kenney, M., Massini, S., Piscitello, L., 2021. Digital technologies, innovation, and skills: emerging trajectories and challenges. Res. Policy 50, 104289. Cohen, S.L., Tripsas, M., 2018, Managing technological transitions by building bridges. Acad. Manag, J. 61, 2319-2342.

Currie, G., Black, J., Duffield, C., 2021. Sustainability assessment of two Australian hydro megaprojects. J. Mega Infrastruct. Sustain. Dev. 1, 255–280. https://doi. org/10.1080/24724718.2021.1930970.

Dainty, A., Leiringer, R., Fernie, S., Harty, C., 2017. BIM and the small construction firm: a critical perspective. Build. Res. Inf. 45, 1–14.

Davies, A., Brady, T., 2000. Organisational capabilities and learning in complex product systems: towards repeatable solutions. Res. Policy 29, 931-953.

Davies, A., Dodgson, M., Gann, D., 2016. Dynamic capabilities in complex projects: the case of London Heathrow Terminal 5. Proj. Manag. J. 47, 26-46.

Davies, A., Gann, D., Douglas, T., 2009. Innovation in megaprojects: systems integration at London Heathrow Terminal 5. Calif. Manage. Rev. 51, 101-125.

Davies, A., MacAulay, S., DeBarro, T., Thurston, M., 2014. Making innovation happen in a megaproject: London's Crossrail suburban railway system. Proj. Manag. J. 45, 25-37.

Davies, A., Mackenzie, I., 2014. Project complexity and systems integration: constructing the London 2012 Olympics and Paralympics Games. Int. J. Project Manage. 32 773-790

De Wilde, P., 2014. The gap between predicted and measured energy performance of buildings: a framework for investigation. Autom. Constr. 41, 40-49.

DeBarro, T., Macaulay, S., Davies, A., Wolstenholme, A., Gann, D., Pelton, J., 2015. Mantra to method: lessons from managing innovation on Crossrail, UK, in: Proceedings of the Institution of Civil Engineers-Civil Engineering. pp. 171-178.

DeFillippi, R.J., Arthur, M.B., 1998. Paradox in project-based enterprise: the case of film making. Calif. Manage. Rev. 40, 125–139.

Denicol, J., Davies, A., Pryke, S., 2021. The organisational architecture of megaprojects. The organisational architecture of megaprojects. Int. J. Project Manage. 39, 339-350.

Denzin, N.K., Lincoln, Y.S., 2017. The SAGE Handbook of Qualitative Research. SAGE Publications, Thousand Oaks.

Diaz, H., Guedes Soares, C., 2020. Review of the current status, technology and future trends of offshore wind farms. Ocean Eng. 209, 10738.

Dille, T., Söderlund, J., 2011. Managing inter-institutional projects: the significance of isochronism, timing norms and temporal misfits. Int. J. Project Manag. 29, 480-490

DiMaggio, P.J., Powell, W.W., 1983. The iron cage revisited: institutional isomorphism and collective rationality in organizational fields. Am. Sociol. Rev. 147–160. Dodgson, M., Gann, D., Macaulay, S., Davies, A., 2015. Innovation strategy in new transportation systems: the case of Crossrail. Transp. Res. Part A 77, 261–275. East, B., Smith, D., 2016. The United States national building information modeling standard. In: The First Decade. 33rd CIB W78 Information Technology for Construction Conference (CIB W78 2016). Brisbane, Australia.

Eastman, C., 1999. Building Product Models: Computer Environments. Supporting Design and Construction. CRC Press, Boca Raton, Florida, USA.

Egan, J., 1998. Rethinking Construction: Report of the Construction Task Force. HMSO, London, UK.

Eggers, J.P., Kaplan, S., 2013. Cognition and capabilities: a multi-level perspective. Acad. Manag. Ann. 7, 295-340.

Eggers, J.P., Kaplan, S., 2009. Cognition and renewal: comparing CEO and organizational effects on incumbent adaptation to technical change. Organ. Sci. 20, 461-477

Eisenhardt, K.M., 1989. Building theories from case study research. Acad. Manag. Rev. 14, 532-550.

Eisenhardt, K.M., Graebner, M.E., 2007. Theory building from cases: opportunities and challenges. Acad. Manag. J. 50, 25-32.

Engwall, M., 2003. No project is an island: linking projects to history and context. Res. Policy 32, 789-808.

Esposito, G., Nelson, T., Ferlie, E., Crutzen, N., 2021. The institutional shaping of global megaprojects: the case of the Lyon-Turin high-speed railway. Int. J. Project Manage. 39, 658-671.

Farjoun, M., 2010. Beyond dualism: stability and change as a duality. Acad. Manage. Rev. 35, 202-225. Farmer, M., 2016. Modernise or Die: The Framer Review of the UK Construction Labour Market. Construction Leadership Council, London.

Fleck, J., 1994. Learning by trying: the implementation of configurational technology. Res. Policy 23, 637-652.

Floricel, S., Bonneau, C., Aubry, M., Sergi, V., 2014. Extending project management research: insights from social theories. Int. J. Project Manage. 32, 1091–1107. Flyvbjerg, B., 2017. The Oxford Handbook of Megaproject Management. Oxford University Press.

² Personal communication

Flyvbjerg, B., 2016. The fallacy of beneficial ignorance: a test of Hirschman's hiding hand. World Dev. 84, 176–189. https://doi.org/10.1016/j. worlddev.2016.03.012.

Flyvbjerg, B., 2014. What you should know about megaprojects and why: an overview. Proj. Manag. J. 45, 6–19. https://doi.org/10.1002/pmj.21409. Flyvbjerg, B., Bruzelius, N., Rothengatter, W., 2003. Megaprojects and Risk: An Anatomy of Ambition. Cambridge University Press, Cambridge, U.K.

Fuenfschilling, L., Truffer, B., 2014. The structuration of socio-technical regimes-Conceptual foundations from institutional theory. Res. Policy 43, 772–791. Gann, D.M., Salter, A.J., 2000. Innovation in project-based, service-enhanced firms: the construction of complex products and systems. Res. Policy 29, 955–972.

Gareis, R., 2010. Changes of organizations by projects. Int. J. Project Manage. 28, 314-327.

Gareis, R., Huemann, M., 2008. Change management and projects. Int. J. Project Manage. 26, 771-772.

Gaunt, M., 2017. BIM model-based design delivery: Tideway East, England, UK. Proceedings of the Institution of Civil Engineers-Smart Infrastructure and Construction 170, 50-58.

GCCG, 2011. Government Construction Client Group. BIM Working Party Strategy Paper, London, UK.

Geels, F., Kern, F., Fuchs, G., Hinderer, N., Kungl, G., Mylan, J., Neukirch, M., Wassermann, S., 2016. The enactment of socio-technical transition pathways: a reformulated typology and a comparative multi-level analysis of the German and UK low-carbon electricity transitions (1990-2014). Res. Policy 45, 896–913. https://doi.org/10.1016/j.respol.2016.01.015.

Geels, F.W., 2020. Micro-foundations of the multi-level perspective on socio-technical transitions: developing a multi-dimensional model of agency through crossovers between social constructivism, evolutionary economics and neo-institutional theory. Technol. Forecast. Soc. Change 152, 119894.

Geels, F.W., 2011. The multi-level perspective on sustainability transitions: responses to seen criticisms. Environ. Innov. Soc. Transit. 1, 24-40.

Geels, F.W., 2010. Ontologies, socio-technical transitions (to sustainability), and the multi-level perspective. Res. Policy 39, 495-510.

Geels, F.W., 2004. From sectoral systems of innovation to socio-technical systems: insights about dynamics and change from sociology and institutional theory. Res. Policy 33, 897–920.

Geels, F.W., 2002. Technological transitions as evolutionary reconfiguration processes: a multi-level perspective and case study. Res. Policy 31, 1257–1274. Geels, F.W., Deuten, J.J., 2006. Local and global dynamics in technological development: a socio-cognitive perspective on knowledge flows and lessons from reinforced concrete. Sci. Public Policy 43, 265–275.

Geels, F.W., Raven, R., 2006. Non-linearity and expectations in niche-development trajectories: ups and downs in Dutch biogas development (1973-2003). Technol. Anal. Strategic Manag. 18, 375–392.

Geels, F.W., Schot, J., 2007. Typology of sociotechnical transition pathways. Res. Policy 36, 399-417. https://doi.org/10.1016/j.respol.2007.01.003.

Geels, F.W., Sovacool, B.K., Schwanen, T., Sorrell, S., 2017. Sociotechnical transitions for deep decarbonization. Science 357, 1242–1244. https://doi.org/10.1126/ science.aao3760.

Genus, A., 1997. Managing large-scale technology and inter-organizational relations: the case of the Channel Tunnel. Res. Policy 26, 169–189.

George, A.L., Bennett, A., 2004. Case Studies and Theory Development in the Social Sciences. MIT Press, Cambridge, MA.

- Gil, N., Miozzo, M., Massini, S., 2012. The innovation potential of new infrastructure development: an empirical study of Heathrow airport's T5 project. Res. Policy 41 (2), 452–466.
- Gil, N.A., Pinto, J., 2016. Collective Action at the Complex Systems Project Front-end: Governance and Performance Implications. Academy of Management, Proceedings. Academy of Management, p. 17194.

Grabher, G., 2004. Temporary architectures of learning: knowledge governance in project ecologies. Organ. Stud. 25, 1491–1514.

Griffin, L.J., 1993. Narrative, event-structure analysis, and causal interpretation in historical sociology. Am. J. Sociol. 98, 1094–1133. https://doi.org/10.1086/ 230140.

Harty, C., 2005. Innovation in construction: a sociology of technology approach. Build. Res. Inf. 33, 512-522.

Harty, C., Whyte, J., 2010. Emerging hybrid practices in construction design work: role of mixed media. J. Constr. Eng. Manag. 136, 468–476.

Hill, C.W.L., Rothaermel, F.T., 2003. The performance of incumbent firms in the face of radical technological innovation. Acad. Manage. Rev. 28, 257-274.

Hobday, M., 2000. The project-based organisation: an ideal form for managing complex products and systems? Res. Policy 29, 871–893.

Holm, P., 1995. The dynamics of institutionalization: transformation processes in Norwegian fisheries. Adm. Sci. Q. 40, 398-422.

Hoogma, R., Kemp, R., Schot, J., Truffer, B., 2002. Experimenting For Sustainable Transport: The Approach of Strategic Niche Management, 1st ed. Routledge, London.

Hornstein, H.A., 2015. The integration of project management and organizational change management is now a necessity. Int. J. Project Manage. 33, 291–298. HS2, 2019. BIM Upskilling Platform [WWW Document]. URL. https://www.bimupskilling.com.

HS2. 2021. High Speed 2 [Online]. Available from: https://www.hs2.org.uk/.

Jick, T., 1979. Mixing qualitative and quantitative methods: triangulation in action. Adm. Sci. Q. 24, 602-611.

Johnson, B.R., Russo, F., Shoonenboom, J., 2017. Causation in mixed methods research: the meeting of philosophy, science and practice. J. Mix Methods Res. 13, 143–162.

Kanger, L., Sovacool, B.K., Noorkõiv, M., 2020. Six policy intervention points for sustainability transitions: a conceptual framework and a systematic literature review. Res. Policy 49.

Keegan, A., Turner, J.R., 2002. The management of innovation in project-based firms. Long Range Plann. 35, 367-388.

Keegan, A., Turner, J.R., 2001. Quantity versus quality in project-based learning practices. Manag. Learn. 32, 77–98.

Kemp, R., Schot, J., Hoogma, R., 1998. Regime shifts to sustainability through processes of niche formation: the approach of strategic niche management. Technol. Anal. Strat. Manag. 10, 175–198. https://doi.org/10.1080/09537329808524310.

Kivimaa, P., Boon, W., Hyysalo, S., Klerkx, L., 2018. Towards a typology of intermediaries in sustainability transitions: a systematic review and a research agenda. Res. Policy 48, 1062–1075.

Klepper, S., 1997. Industry life cycles. Ind. Corp. Change 6, 145–182.

Klepper, S., 1996. Entry, exit, growth and innovation over the product life cycle. Am. Econ. Rev. 86, 560-581.

Köhler, J., Geels, F.W., Kern, F., Markard, J., Wieczorek, A., Alkemade, F., Avelino, F., Bergek, A., Boons, F., Fünfschilling, L., Hess, D., Holtz, G., Hyysalo, S., Jenkins, K., Kivimaa, P., Martiskainen, M., McMeekin, A., Mühlemeier, M.S., Nykvist, B., Onsongo, A., Pel, B., Raven, R., Rohracher, H., Sandén, B., Schot, J., Sovacool, B., Turnheim, B., Welch, D., Wells, P., 2019. An agenda for sustainability transitions research: state of the art and future directions. Environ. Innov. Soc. Transit. 31, 1–32.

Konrad, K., 2006. The social dynamics of expectations: the interaction of collective and actor-specific expectations on electronic commerce and interactive television. Technol. Anal. Strat. Manag. 18, 429–444. https://doi.org/10.1080/09537320600777192.

Koskinen, K.U., 2008. Boundary brokering as a promoting factor in competence sharing in a project work context. Int. J. 1, 119–132.

Koskinen, K.U., Pihlanto, P., Vanharanta, H., 2003. Tacit knowledge acquisition and sharing in a project work context. Int. J. Project Manage. 21, 281–290. Langley, A., 1999. Strategies for theorizing from process data. Acad. Manage. Rev. 24, 691–710.

Langley, A., Ravasi, D., 2019. Visual artifacts as tools for analysis and theorizing, in: Zilber T.B., A.J.M.M.J. (Ed.), The Production of Managerial Knowledge and Organizational Theory: New Approaches to Writing, Producing and Consuming Theory Research in the Sociology of Organizations Vol. 59. Emerald Publishing Limited, pp. 173–199.

Langley, A., Smallman, C., Tsoukas, H., de Ven, A.H., 2013. Process studies of change in organization and management: unveiling temporality, activity, and flow. Acad. Manag. J. 56, 1–13. https://doi.org/10.5465/amj.2013.4001.

Latham, M., 1994. Constructing the Team: Joint review of Procurement and Contractual Arrangements in the United Kingdom Construction Industry. HMSO, London. Lavie, D., 2006. Capability reconfiguration: an analysis of incumbent responses to technological change. Acad. Manage. Rev. 31, 153–174.

Leonard-Barton, D., 1988. Implementation as mutual adaptation of technology and organisation. Res. Policy 17, 251–267.

Levina, N., Vaast, E., 2005. The emergence of boundary spanning competence in practice: implications for implementation and use of information systems. MIS quarterly 29, 335-363.

Levinthal, D.A., 1998. The slow pace of rapid technological change: gradualism and punctuation in technological change. Ind. Corp. Change 7, 217–246. Levy, J.S., 2008. Case studies: types, designs, and logics of inference. Confl. Manag. Peace Sci. 25, 1–18.

Llewelyn, A.I., 1989. Review of CAD/CAM. Comput. Aided Des. 21, 297-302.

Lundin, R.A., Arvidsson, N., Brady, T., Ekstedt, E., Midler, C., Sydow, J., 2015. Managing and Working in Project Society: Institutional Challenges of Temporary Organizations. Cambridge University Press, Cambridge, UK.

Malerba, F., Orsenigo, L., 1996. Schumpeterian patterns of innovation are technology-specific. Res. Policy 25, 451–478.

Manning, S., 2008. Embedding projects in multiple contexts - a structuration perspective. Int. J. Project Manage. 26, 30-37.

Manning, S., Sydow, J., 2011. Projects, paths, and practices: sustaining and leveraging project-based relationships. Ind. Corp. Chang. 20 (5), 1369–1402.

Martinsuo, M., Hoverfält, P., 2018. Change program management: toward a capability for managing value-oriented, integrated multi-project change in its context. Int. J. Project Manage. 36, 134–146.

Matten, D., Crane, A., 2005. Corporate citizenship: toward an extended theoretical conceptualization. Acad. Manag. J. 30, 166-179.

McKelvey, B., 1982. Organizational Systematics: Taxonomy, Evolution, Classification. University of California Press, Berkeley, CA.

Menezes, A.C., Cripps, A., Bouchlaghem, D., Buswell, R., 2012. Predicted vs. actual energy performance of non-domestic buildings: using post-occupancy evaluation data to reduce the performance gap. Appl. Energy 97, 355–364.

Midler, C., 1995. Projectification" of the firm: the Renault case. Scand. J. Manag. 11, 363-375.

Milliken, F.J., 1987. Three types of perceived uncertainty about the environment: state, effect, and response uncertainty. Acad. Manage. Rev. 12, 133–143. Morgan, B., 2020. Tideway's digital evolution: using digital technology to drive productivity gains.

Morgan, B., 2019. Organizing for digitalization through mutual constitution: the case of a design firm. Construct. Manag. Economics. 37, 400-417.

Morgan, B., 2017. Organizing for digitization in firms: a multiple level perspective. In: Chan, P.W., Neilson, C.J. (Eds.), Proceedings of the 33RD Annual Association of Researchers in Construction Management Conference (ARCOM 2017). Association of Researchers in Construction Management, Cambridge, UK, pp. 4–6.

Morris, P.W.G., 2013. Reconstructing Project Management. John Wiley & Sons.

Morris, P.W.G., 2012. Cleland and King: project management and the systems approach. Int. J. Manag. Projects Bus. 5, 634-642.

Morris, P.W.G., Geraldi, J., 2011. Managing the institutional context for projects. Proj. Manag. J. 42, 20-32.

Müller, R., Zhai, L., Wang, A., Shao, J., 2016. A framework for governance of projects: governmentality, governance structure and projectification. Int. J. Project Manage. 34, 957–969.

Munsi, A., 2012. Crossrail and Bentley Systems Launch UK's First Dedicated Building Information Modelling Academy.

Office, C., 2016. Government Construction Strategy: 2016 - 2020. Cabinet Office and Infrastructure and Projects Authority, UK [Online]Available: [Accessed], London.

Orlikowski, W.J., 1992. The duality of technology: rethinking the concept of technology in organizations. Organ. Sci. 3, 398-427.

Oudshoorn, N., Pinch, T. (Eds.), 2003. How Users Matter: The Co-Construction of Users and Technology. MIT Press, Cambridge, MA.

Papachristos, G., Sofianos, A., Adamides, E., 2013. System interactions in socio-technical transitions: extending the multi-level perspective. Environ. Innov. Soc. Transit. 7 https://doi.org/10.1016/j.eist.2013.03.002.

Papadonikolaki, E., van Oel, C., Kagioglou, M., 2019. Organising and Managing boundaries: a structurational view of collaboration with Building Information Modelling (BIM). Int. J. Project Manage. 37 (3), 378–394.

Papadonikolaki, E., Wamelink, H., 2017. Inter-and intra-organizational conditions for supply chain integration with BIM. Build. Res. Inf. 45 (6), 649–664. Pavitt, K., 1984. Sectoral patterns of technical change: towards a taxonomy and a theory. Res. Policy 13, 343–373.

Peansupap, V., Walker, D.H.T., 2007. Innovation diffusion at the implementation stage of a construction project: a case study of information communication technology. Construct. Manag. Econ. 24, 321–332.

Pelton, J., Brown, M., Reddaway, W., Gilmour, M., Phoon, S., Wolstenholme, A., Gann, D.C., 2017. The evolution of an innovation ecosystem. In: Proceedings of the Institution of Civil Engineering. Thomas Telford Ltd, pp. 181–190.

Pettigrew, A.M., 1990. Longitudinal field research on change: theory and practice. Organ. Sci. 1, 267-292.

Podkolinski, J., 2021. HS2 railway, UK-the hybrid bill. In: Proceedings of the Institution of Civil Engineers-Transport, 174. Thomas Telford Ltd., pp. 59–66 Pollalis, S.N., Georgoulias, A., 2008. Channel Tunnel Rail Link Risk Transfer and Innovation in Project Delivery. Harvard Design School, Boston, MA, US. Poole, M.S., van de Ven, A.H., 1989. Using paradox to build management and organization theories. Acad. Manage. Rev. 14, 562–578.

Pöttler, R., 1992. Three-dimensional modelling of junctions at the Channel Tunnel project. Int. J. Numer. Anal. Methods Geomech. 16, 683-695.

Prencipe, A., Tell, F., 2001. Inter-project learning: processes and outcomes of knowledge codification in project-based firms. Res. Policy 30 (9), 1373–1394.

Qiu, Y., Chen, H., Sheng, Z., Cheng, S., 2019. governance of institutional complexity in megaproject organizations. Int. J. Project Manage. 37, 425–443.

Raven, R., 2007. Co-evolution of waste and electricity regimes: multi-regime dynamics in the Netherlands (1969-2003). Energy Policy 35, 2197–2208.

Raven, R., Geels, F.W., 2010. Socio-cognitive evolution in niche development: comparative analysis of biogas development in Denmark and the Netherlands (1973-2004). Technovation 30, 87-99.

Reichstein, T., Salter, A.J., Gann, D.M., 2005. Last among equals: a comparison of innovation in construction, services and manufacturing in the UK. Construct. Manag. Econ. 23, 631–644.

Rip, A., Kemp, R., 1998. Technological change. In: Rayner, S., E.L.M (Eds.), Technological change. Human Choice and Climate Change Vol. 2 327–399. Rosenberg, N., 1976. On Technological Expectations. Econ. J. 86, 523–535. https://doi.org/10.2307/2230797.

Rosenbloom, D., 2020. Engaging with multi-system interactions in sustainability transitions: a comment on the transitions research agenda. Environ. Innov. Soc. Transit. 34, 336-340.

Saldana, J., 2009. The Coding Manual For Qualitative Researchers. Sage Publications, Los Angeles, CA.

Sarantakos, S., 2005. Social Research.

Schmidt, S., Werle, R., 1998. Coordinating Technology: Studies in the International Standardization of Telecommunications. MIT Press, Cambridge MA.

Schot, J., Geels, F.W., 2008. Strategic niche management and sustainable innovation journeys: theory, findings, Research agenda, and policy. Technol. Anal. Strat. Manag. 20, 537–554.

Schot, J., Geels, F.W., 2007. Niches in evolutionary theories of technical change. J. Evol. Econ. 17, 605-622.

Sense, A.J., Antoni, M., 2003. Exploring the politics of project learning. Int. J. Project Manage. 21, 487–494.

Sergeeva, N., Winch, G.M., 2020. Narrative interactions: how project-based firms respond to government narratives of innovation. Int. J. Project Manage. 38, 379–387.

Shibeika, A., Harty, C., 2015. Diffusion of digital innovation in construction: a case study of a UK engineering firm. Construct. Manag. Econ. 33, 453-466.

Smith, A., Raven, R., 2012. What is protective space? Reconsidering niches in transitions to sustainability. Res. Policy 41, 1025–1036.

Smith, A., Voss, J.-.P., Grin, J., 2010. Innovation studies and sustainability transitions: the allure of the multi-level perspective and its challenges. Res. Policy 39, 435–448. https://doi.org/10.1016/j.respol.2010.01.023.

Söderlund, J., 2011. Pluralism in project management: navigating the crossroads of specialization and fragmentation. Int. J. Manag. Rev. 13, 153–176. Söderlund, J., Sydow, J., 2019. Projects and institutions: towards understanding their mutual constitution and dynamics. Int. J. Project Manage. 37, 259–268.

Sovacool, B.K., Geels, F.W., 2021. Megaprojects: Examining their Governance and Sociotechnical Transitions Dynamics. Environmental Innovation and Societal Transitions In Press. Corrected Proof.

Suarez, F.F., Utterback, J.M., 1995. Dominant designs and the survival of firms. Strat. Manag. J. 16, 415-430.

Sutton, R.I., Staw, B.M., 1995. What theory is not. Adm. Sci. Q. 40, 371-384.

Swedberg, R., 2017. Theorizing in sociological research: a new perspective, a new departure? Annu. Rev. Sociol. 42, 189-206.

Swedberg, R., 2016. Can you visualize theory? On the use of visual thinking in theory pictures, theorizing diagrams and visual sketches. Sociol. Theory 34, 250–275. Sydow, J., Braun, T., 2018. Projects as temporary organizations: an agenda for further theorizing the interorganizational dimension. Int. J. Project Manage. 36, 4–11. Sydow, J., Lindkvist, L., DeFillippi, R., 2004. Project-based organizations, embeddedness and repositories of knowledge: editorial. Organ Stud 25, 1475–1489.

Sydow, J., Windeler, A., 2020. Temporary organizing and permanent contexts. Curr. Sociol. 68, 480-498.

Thiel, J., Grabher, G., 2015. Crossing boundaries: exploring the London Olympics 2012 as a field-configuring event. Ind. Innov. 22, 229-249.

Tideway, 2021. Thames Tideway Tunnel: London's Super Sewer [Online]. Available: [Accessed].

Too, E.G., Weaver, P., 2014. The management of project management: a conceptual framework for project governance. Int. J. Project Manage. 32, 1382–1394. Tripsas, M., Gavetti, G., 2000. Capabilities, cognition, and inertia: evidence from digital imaging. Strat. Manag. J. 21, 1147–1161.

Turner, J.R., 1999. The Handbook of Project-Based Management, 2nd edition. McGraw-Hill, Maidenhead.

Turnheim, B., Geels, F.W., 2019. Incumbent actors, guided search paths, and landmark projects in infrasystem transitions: *re*-thinking Strategic Niche Management with a case study of French tramway diffusion (1971–2016). Res. Policy 48, 1412–1428.

Tyre, M.J., Orlikowski, W.J., 1994. Windows of opportunity: temporal patterns of technological adaptation in organizations. Organ. Sci. 5, 98-117.

Zerjav, V., McArthur, J., Edkins, A., 2021. The multiplicity of value in the front-end of projects: the case of London transportation infrastructure. Int. J. Project Manage. 39, 507–519.

Von Hippel, R., 1986. Lead users: a source of novel product concepts. Manage. Sci. 32, 791-805.

- Vuorinen, L., Martinsuo, M., 2018. Program integration in multi project change programs: agency in integration practice. Int. J. Project Manage. 36, 583-599.
- Weick, K.E., 1995. What theory is not, theorising is. Adm. Sci. Q. 40, 385-390.

Whitley, R., 2006. Project-based firms: new organizational form or variations on a theme? Ind. Corp. Change 15, 77-99.

Whyte, J., 2019. How digital information transforms project delivery models. Proj. Manag. J. 50, 177-194.

Williams, T.M., 2008. How do organizations learn lessons from projects-and do they? IEEE Trans. Eng. Manage. 55, 248-260.

Winch, G., 1998. Zephyrs of creative destruction: understanding the management of innovation in construction. Build. Res. Inf. 26, 268-279.

Winch, G.M., Maytorena-Sanchez, E., 2020. Institutional projects and contradictory logics: responding to complexity in institutional field change. Int. J. Project Manage. 38, 368–378.

Windeler, A., Sydow, J., 2001. Project networks and changing industry practices: collaborative content production in the {G}erman television industry. Organ. Stud. 22, 1035–1060.

Woetzel, J., Garemo, N., Mischke, J., Hjerpe, M., Palter, R., 2016. Bridging Global Infrastructure Gaps. www.mckinsey.com/industries/infrastructure/our-insights/ bridging-global-infrastructure-gaps.

Wolstenholme, A., Austin, S.A., Bairstow, M., Blumenthal, A., Lorimer, J., Mcguckin, S., Jones, S.R., Ward, D., Whysall, D., Le Grand, Z., Guthrie, W., Davies, R., 2009. Never Waste a Good crisis: a Review of Progress Since Rethinking Construction and Thoughts For Our Future. Constructing Excellence, London.

Yin, R.K., 1984. Case Study Research: Design and Methods. SAGE Publications, Beverly Hills, California.